Author’s response to reviews

Title: Collaborative development of an electronic Personal Health Record for people with severe and enduring mental health problems

Authors:

Liam Ennis (liam.ennis@kcl.ac.uk)
Daniel Robotham (Dan.Robotham@kcl.ac.uk)
Mike Denis (mike.denis@medsci.ox.ac.uk)
Ninjeri Pandit (Ninjeri.Pandit@nhs.net)
Dave Newton (David.Newton@slam.nhs.uk)
Diana Rose (Diana.Rose@kcl.ac.uk)
Til Wykes (Til.wykes@kcl.ac.uk)

Version: 4 Date: 13 October 2014

Author’s response to reviews:

We have placed the table at the end of the document. We have added to the methods section the detail that participants provided written informed consent (p4). Email addresses have been added to the title page of the manuscript.

Many thanks for taking the time to review our manuscript.

Many thanks for taking the time to review our paper.